THE IMPACT OF JUDICIAL CONSTITUTIONAL COURT REVIEW OF LEGISLATIVE OMISSION ON CONSTITUTIONAL AUTHORITARIANISM

Karem Sayed Aboelazm *, Raghda Raafat **, Hanadi Sharif ***, Emad Ibrahim ****

* College of Law, United Arab Emirates University, Abu Dhabi, UAE
Contact details: College of Law, United Arab Emirates University, P. O. Box 15551, Abu Dhabi, UAE

** College of Law, City University of Ajman, Ajman, UAE

*** College of Law, University of Khorfakkan, Sharjah, UAE

**** College of Law, Umm Al Quwain University, Umm Al Quwain, UAE



How to cite this paper: Aboelazm, K. S., Raafat, R., Sharif, H., & Ibrahim, E. (2025). The impact of judicial constitutional court review of legislative omission on constitutional authoritarianism. Corporate Law & Governance Review, 7(4), 17–29. https://doi.org/10.22495/clgrv7i4p2

Copyright © 2025 The Authors

This work is licensed under a Creative Commons Attribution 4.0 International License (CC BY 4.0). https://creativecommons.org/licenses/by/4.0

ISSN Online: 2664-1542 ISSN Print: 2707-1111

Received: 11.03.2025

Revised: 27.06.2025; 10.09.2025; 19.11.2025

19.11.2025

Accepted: 24.11.2025

JEL Classification: K10, K15, K19, K24 **DOI:** 10.22495/clgrv7i4p2

Abstract

This paper aims to provide an analysis of the impact of constitutional review on legislative omissions, their impact on constitutional authoritarianism by the Supreme Constitutional Court, and how constitutional review affects rights and freedoms, as well as the validity of legislation. The paper also aims to present the effects of constitutional review on the relationship between authorities (Al-Dughili, 2018). In light of the principle of separation of powers, and the potential for tension between the judiciary and the legislature. The paper employs an analytical approach to examine judicial rulings issued by the Supreme Constitutional Court, aiming to determine the criteria for ruling on the unconstitutionality of legislative omissions, the extent of constitutional judicial intervention in the legislative process, and the potential repercussions of such rulings on all aspects. The paper reached several conclusions, the most important of which is that the Supreme Constitutional Court, with its authority to exercise constitutional oversight over legislative actions, can maintain legal security. However, constitutional judicial oversight over legislative omissions has some adverse effects, which are represented by tension in the relationship between the constitutional judiciary and the legislative authority, which may lead to exposing the powers of the constitutional judiciary to constitutional judicial oversight over legislative omissions to danger. It may also contribute to legislative tyranny and the seizure of legislative authority by the constitutional judiciary.

Keywords: Constitutional Court, Judicial Review, Constitutionality, Legislative Omission

Author's individual contribution: Conceptualization — K.S.A., R.R., H.S., and E.I.; Methodology — K.S.A., R.R., H.S., and E.I.; Writing — Original Draft — K.S.A., R.R., H.S., and E.I.; Writing — Review & Editing — K.S.A.; Supervision — K.S.A.; Project Administration — K.S.A.

Declaration of conflicting interests: The Authors declare that there is no conflict of interest.

1. INTRODUCTION

The effects of judicial oversight over legislative omission vary depending on the extent to which the constitutional judge adheres to the limits and controls of this oversight (Al-Aboudi, 2010). Suppose the constitutional courts adhere to the limits and controls of exercising this oversight. In that case, they will become an effective tool for confronting the negativity of the legislative authority, preserving



the quality and integrity of legislation. They will also become effective in fostering a democratic dialogue between the constitutional judiciary and the three branches of government, comprising legislative, executive, and judicial authorities (Al-Dughili, 2018). However, if the constitutional judge exceeds the limits of his oversight authority and its controls, this would be enough to cause several imbalances in the entire legal system of the state, which leads relationship to tension in the between the constitutional oversight authority and other public authorities, especially the legislative authority, which may undermine seek to the jurisdiction of the constitutional judiciary, which ultimately reflects negatively on individual rights and freedoms (Abdullah, 2023; Ali, 1999).

Therefore, constitutional judicial oversight of legislative omission remains a matter of judicial jurisprudence that appears and recedes according to the circumstances and will continue to do so until a legislative text explicitly approves it (Mohammed, 2019; Al-Sanhouri, 1992). The emergence and decline of constitutional judicial oversight over legislative to the omissions are due alertness the constitutional judge's conscience and its decline, as well as the strength of the judicial body that undertakes oversight of constitutionality. Alternatively, its disintegration and the diminishing of its role between the will to develop the constitution and the warnings of deviating from it, even if that is indirect (Shaban, 2021;

Although all the jurisprudential attempts made so far have been unable to adequately address the mechanism of constitutional judicial oversight of legislative omissions, it cannot be denied that constitutional judicial oversight of legislative omissions creates a unique dynamic for countries that aspire to approach the model of continuous participatory democracy (Aboelazm, Al-Shennawy, 2017). It is also a tool for balancing public authorities, facilitating democratic dialogue between the constitutional judiciary and other public authorities, and promoting the stability of legal relations. It also ensures that the legislative authority is held accountable for failing to activate constitutionally stipulated rights and freedoms. Additionally, it serves as a tool for maintaining the quality and integrity of legislation (Al-Shimi, 2003).

The exercise of constitutional judicial oversight over legislative omissions is surrounded by many concerns, given that the judge's overstepping in exercising that oversight will significantly weaken societal confidence in the rulings of the constitutional judiciary, in addition to the legislative authority lurking in it to undermine and limit its jurisdiction, which is what. It will lead to the relationship between the constitutional judiciary and the legislative authority becoming increasingly tense and conflicting (Aboelazm, 2024b).

Accordingly, the study will examine the positive and negative effects of constitutional judicial oversight on legislative negligence by presenting the impact of constitutional judicial oversight on legislative negligence regarding the protection of rights and freedoms, and also regarding the quality and integrity of legislation, as well as its impact on the relationship between the constitutional judiciary and the legislative authority, and finally, the effects related to the jurisdiction of the constitutional judiciary (Abu Halima, 2015).

Furthermore, constitutional judicial oversight of legislative omissions produces numerous effects and repercussions on multiple parties, some of which relate to individual rights and freedoms. In contrast, others relate to the integrity and quality of legislation. This is in addition to the effects it has on the relationship between the legislative authority and the constitutional judiciary, as well as the impact of this oversight on the powers of the constitutional judiciary, which results from its exercise of this type of oversight. Accordingly, the research problem is represented in the following question: What are the repercussions and effects of constitutional oversight of legislative omission?

Several other research questions branch out from the main research question, which are as follows:

RQ1: What are the effects of constitutional judicial oversight on legislative omissions regarding the constitutional protection of rights and freedoms?

RQ2: What are the effects of constitutional judicial oversight on legislative omissions regarding the quality and integrity of legislation?

RQ3: What are the effects of constitutional judicial oversight on legislative omissions regarding the relationship between the constitutional judiciary and the legislative authority?

RQ4: What are the effects of constitutional judicial oversight on legislative omissions regarding the jurisdiction of the constitutional judiciary?

The rest of this paper is structured as follows. Section 2 reviews the relevant literature. Section 3 provides the methodology employed in conducting the research. Section 4 presents the study's results. Section 5 concludes the research paper, outlining future research directions.

2. LITERATURE REVIEW

2.1. Legislative omissions vs. human rights

2.1.1. Constitutional judicial oversight of legislative omissions reinforces the principle of legal security

The principle of legal security in society depends on the method chosen by the legislator to organize the protection it adds to rights, freedoms, and the public interest (Bustamante & Bustamante, 2011). This security is available whenever this interest is legally protected in a manner that ensures consistency and stability, and secures citizens' rights and freedoms against unforeseen events that could endanger their legal positions. The balance between the requirements of the public interest and the protection of rights and liberties is considered the primary guarantor of legal security, in addition to being the responsibility of the legislative authority (Nassar, 2011).

According to the doctrine of practical philosophy, the function of law is to protect public rights and freedoms. The state bears the burden of providing security, justice, and reassurance, coordinating individual activities in a way that prevents them from conflicting, and satisfying the basic needs of individuals (Zahra, 2013). Therefore, inadequate regulation of these issues or neglecting certain aspects would have a severe impact on society. Thus, constitutional oversight balanced he compatible and must considerations of legal security by choosing a solution that does not result in a legislative vacuum or other consequences that could sacrifice legal stability in the country (Aboelazm, 2025; Idris, 2011).

The primary goals and objectives of legislation are to maintain security and order in social, political, and economic life and to ensure the stability of relations and transactions within The existence of the law, the quality of its rules, the accuracy of its formulation, and the treatment of cases for which it was intended to regulate are what achieve that goal (El-Sawy, 2003). It ensures the regularity of society's political, social, and cultural life. Additionally, this enhances security in individuals' relationships with one another. There is no doubt that inadequate regulation of a particular legislative issue, or neglecting its regulation despite its importance, leads to societal destabilization (Csink & Pasczolay, 2008).

The Constitutional Court, in its role within system, must exercise its effectively amid a network of conflicting challenges (Belgilaly, 2017b). Among these challenges is the stability of the contested legislative texts for some to a according meaning, time specific the communication of everyone's knowledge with this meaning, and the arrangement of their affairs accordingly. This creates a fence of legal security that revolves around this meaning. Therefore, ruling these texts completely unconstitutional yields unfortunate consequences regarding the principle of legal protection. The principle serves as the basis for citizens' fundamental rights freedoms, and is a significant element in establishing the rule of law (Blondio-Mondoloni, 2014).

Accordingly, legislative negligence leads to compromising the principle of legal security, which is based on the existence of model legal rules that are easy to understand, access, and apply, and to anticipate their results and effects in a way that guarantees the stability of legal relations between individuals and between them and the state (Al-Ghafloul, 2003). This is because the legislative omission contradicts the legitimate expectations of individuals, given that the legislator did not cover all aspects of the subject being regulated; that is, it approved the right or freedom being regulated for one group over another, despite the similar legal status of each (Aboelazm, 2024a).

Therefore, the constitutional judge's approach to abstract oversight does not suit the case of legislative negligence. On the one hand, canceling the right or benefit granted by the legislator through the contested text harms the group for whose benefit it was decided (Al-Aboudi, 2010). On the other hand, repealing the legislative text does not guarantee complete protection of the principle of equality. At the same time, a particular group may enjoy the right or benefit decided by the legislator during the period preceding the issuance of the ruling or constitutional decision to cancel the legislative text assigned to them to exclude other groups (Al-Janabi, 2022). On the final point, canceling the text subject to constitutional judicial legislative oversight for omissions harms the appellant, as the invalidity applies to the text entirely, depriving the stipulated and omitted categories of the legally stipulated right, rather than establishing the right by analogy with the specified category (AbdelRahman, 2016).

Therefore, following constitutional judicial oversight mechanisms over legislative omissions ensures that the principle of legal security is not

wasted by avoiding the execution of the text subject to oversight due to omission after it has been declared a right or freedom for a group of individuals (Abdelkarim & El-Emara, 2019: Aboelazm, 2024a). Its application is stabilized for a period during which legitimate legal positions and positions are arranged based on it (Abboud, 2019). Therefore, ruling this text unconstitutional due to its retroactive effect leads to the creation of a legislative vacuum on the one hand and the destabilization of those legal conditions and positions on the other hand, which negatively affects the achievement of the principle of legal security, which requires achieving a degree of consistency and stability of the legal base, and respecting the resulting rights and legal positions of individuals (Abdel-Badi, 2019).

Through mechanisms for monitoring legislative omissions, the constitutional judge can extend the advantages or benefits granted to the category stipulated in the text, subject to oversight of the category or categories that the legislator did not in this text, to the constitutionally established principle of equality. In a manner that preserves the stability of the legal centers and situations, and guarantees the supremacy of constitutional provisions (Aboelazm, 2025). This aims to strike a balance between the effectiveness of constitutional oversight on the one hand and the requirements of legal security on the other by mitigating the judiciary's tendency to overrule total unconstitutionality, which applies retroactively, thereby preventing severe disturbances that violate the principle of legal protection (Al-Dulaimi, 2018).

Therefore, the only most appropriate way before the constitutional judiciary is to restore the principle of equality and avoid the constitutional defect arising from the omission of legislation, which is represented in extending the advantage or benefit to other groups that were overlooked by the contested text, instead of ruling that the text is entirely unconstitutional, which results in a violation of the positions that it has been formed since the issuance of this text (Al-Murr, 2003). From this standpoint, oversight of legislative negligence is necessary, as it contributes to strengthening and developing legal security on the one hand and maintaining the stability of legal institutions on the other (Al-Najjar, 2010).

2.1.2. Legislative omission vs. the integration of authorities

Constitutional judicial oversight of legislative represents development omissions a constitutional oversight. A creative role enables devise the constitutional judge to technical mechanisms for balancing conflicting interests within society (Al-Turkmani, 2022). Although the new mechanisms or approaches to judicially address legislative oversight represent a departure from the usual framework of constitutional oversight, they lead to avoiding many practical ruling problems resulting from unconstitutionality and the resulting instability of legal centers (Brewer-Carías, 2011). It is also a means of dialogue with various political forces and institutions, especially the Parliament the executive authority, to reduce the intensity of political conflicts between them (Haji, 2012).

The new mechanisms for constitutional judicial of legislative omissions the constitutional judge to effectively address this omission, as it is a particular type of constitutional defect, and are more respectful of Parliament's powers than traditional techniques of constitutional oversight (El-Gamal, 2000). The ruling issued on the unconstitutionality of the text subject to censorship in what it did not include or in what it omitted has a retroactive effect and, therefore, entails correcting the defect in the text and restoring the necessary balance between the categories or cases, stipulated and unstated, in a way that is consistent with the legitimate expectations of individuals from On the one hand, it guarantees the preservation of the legislative text issued by the legislative authority on the other hand (Halmai, 2019).

Therefore, oversight of legislative negligence is necessary, as it enables the implementation of constitutionally guaranteed rights and freedoms, particularly those whose effectiveness depends on legislative rules (Melchior & Courtoy, 2007). It also contributes to clarifying how the competent authorities correctly apply legislative texts. Therefore, the intervention of the constitutional courts to fill the legal gaps resulting from legislative negligence creates a kind of democratic dialogue with the legislator by addressing some of the guiding principles that must be adhered to when amending the contested law, which the court ruling saved from the constitutional challenges directed at it (Melchior & Courtoy, 2007).

Allowing constitutional judicial oversight of legislative omissions opens the way for developing mechanisms and outcomes of constitutional oversight (Legislative omission in constitutional jurisprudence, 2008). This is done by avoiding excessive rulings issued that are unconstitutional without wasting the essence of the constitutional judiciary by not allowing the application of texts that contradict the provisions of the Constitution in a way that preserves the balance between the authority of the oversight body and the function of Parliament especially since it does not appear to be the supreme legislator in practice, which would avoid a clash between them because it allows the constitutional judge the authority to revise the legislative text without the need to rule that it be entirely invalidated by giving an interpretation of this text that is consistent with the provisions of the Constitution to preserve it and avoid declaring it unconstitutional (Mohammed, 2019).

The mechanisms developed legislative omissions have enabled the constitutional to establish their relationships with the legislative authority on the one hand and with the judiciary on the other, thereby upholding the constitutional rules within the legal system (Salman, 2021). One of the most important of these mechanisms is the provisions hosting the principles, through which the constitutional judge begins a democratic dialogue not only with the legislative authority called upon to avoid unconstitutional omission but also with the judges of the subject who are entrusted in the event of legislative failure to apply the principle contained in the ruling or decision issued regarding unconstitutionality to the relations or the issues that are presented to them as well (Salman, 2019).

This transformation leads to an active relationship between the constitutional judiciary and

the legislative authority, which merely abolishes laws and directs the legislator to establish some legislative rules (Shaban, 2021). The constitutional judge does not seek to oppose the political authority but instead aims to restore it to the constitutional principles that are binding on all authorities, including the elected authority, thus avoiding a sharp clash with the parliamentary majority that voted in favor of the legislation if the constitutional judiciary were to abolish it altogether. Through this oversight, the constitutional judge can balance respecting the majority's will on the one hand and not violating rights and freedoms on the other hand (Shandy, 2017). There is no doubt that this democratic "legislative-judicial" dialogue contributes to bringing the views of the legislative and judicial authorities closer together, thus avoiding conflicts and clashes between them (Al Dalaien & Aladaseen, 2025; Sólyom, 2003).

Accordingly, the constitutional provisions and decisions that address legislative omissions are a very fertile and rich means of dialogue between the constitutional judiciary on the one hand and legislative authority and the subject judge on the other hand (Shahat, 2004). Examining the Egyptian case, it is evident that the Egyptian legislator's interaction with constitutional rulings declaring legislative omissions unconstitutional appears minimal (Serra et al., 2008). Anyone who follows the rulings issued by the Supreme Constitutional Court in this regard, and the behavior of the legislative authority towards them, can quickly notice the rarity of the legislator's response the constitutional rulings that lead the unconstitutionality of the legislative omission related to the text challenged before the court. After the verdict was issued, the subject court was instructed to apply its rulings to the individual cases presented, and it left the issue. It is necessary, given that constitutional provisions enjoy absolute authority vis-à-vis everyone and all state authorities (Belgilaly, 2017a).

2.1.3. Legislative omission and the unconstitutionality

Constitutional judicial oversight of legislative omissions enables constitutional courts to imbue legal rules with a constitutional character, thereby ensuring the supremacy of constitutional legitimacy (Arlettaz, 2015). Through it, the court can hold the legislator accountable for their legislative responsibilities and enhance the trial judge's role in upholding constitutional values (Azawi, 2011). This oversight helps determine the foundations and controls that must be adhered to by the legislative authority when exercising its powers to develop these powers. The legislative authority cannot avoid the principles and foundations in judicial precedents issued by the constitutional judiciary (El-Din, 1986).

The constitutional oversight body reveals the areas of unconstitutional legislative omissions through its rulings or decisions. It leaves it to the legislative authority to avoid the effects of unconstitutionality related to this oversight (Hasan, 2015). The executive authority and the subject judge are also obligated to implement and apply the legislative text in conjunction with the cases or conditions mentioned by the constitutional oversight body to guarantee its constitutionality in implementing the absolute authority enjoyed by the rulings and decisions issued in constitutional cases (Al-Dughili, 2018). Suppose the legislator does

not intervene to find the legal rule that meets the constitutional requirements. In that case, the trial judge can rely on the "principle" established by the Constitutional Court to deduce the legal rule applicable to the cases presented before him from the legal system (Al-Shennawy, 2017).

Therefore, by using this tool, the constitutional judge can get out of the rigid framework for monitoring the constitutionality of laws, which is represented by deciding the constitutionality or unconstitutionality of the contested legislative text, which leads to not expanding the issuance of rulings of unconstitutionality, biasing the established principle of constitutionality in favor of the legislative text (Al-Janabi, 2022). Furthermore, it avoids a clash between the constitutional judiciary, the state authorities in general, and the legislative authority in particular. This represents a barrier to all attempts to reduce the scope of constitutional oversight in quantity and quality (Al-Aboudi, 2010).

Accordingly, the goal of constitutional judicial oversight of legislative omissions is to evaluate the process of making legislation in a way that makes it more compatible with the overall objectives of the constitutional document "the Constitution", by imposing restrictions or conditions that depend on the constitutionality of the legislative text under oversight to the extent and limits of its respect, instead of resorting to deciding that it is entirely which unconstitutional, leads to the legislative gap contained in the legal system (Abdel-Badi, 2019). Suppose legal motives govern the rulings and decisions of the constitutional courts, and they cannot consider the political, economic, and social aspects except through the principles or provisions contained in the texts. In that constitutional however. case. the contributions of the constitutional judiciary over the years to improving the quality of laws cannot be denied (Abdulkareem & Al-Amarah, 2019).

Therefore, constitutional judicial oversight of legislative omissions is one of the determinants of constitutional oversight (Abboud, 2019). Through it, constitutional judge can reconcile conflicting considerations (Abu Halima, The first is to respect the legislator's will, not cancel the legislative text within the framework of his appreciation of society's need for this text. The second is the necessity of conforming legislative texts to the provisions of the Constitution (Al-Sayed, 2009). The Constitution sets conditions that guarantee the constitutionality of the contested text. This represents a new dimension to the presumption of constitutionality established in the interest of legislation, which constitutional judicial oversight supplements legislative oversight (Al-Ghafloul, 2003).

The bottom line is that constitutional judicial oversight of legislative omission achieves several goals and various benefits for the legal system in a modern democracy (Blondio-Mondoloni, 2014). On the one hand, it allows the constitutional judge to preserve the sovereignty of the Constitution's provisions and constrain them from forms of aggression against these provisions, even if breaching them is masked by a negative guise, such as Parliament's failure to implement these provisions (Halmai, 2019). On the other hand, it enables the constitutional judge to alert Parliament to the need to address unconstitutional legislative omissions and to oblige or force it to address these omissions within a specified period. On the other hand, it opens the way for the constitutional judge

to provide a helping hand to the legislative authority by addressing legislative omissions that contradict the Constitution and constitutional guidelines, and by presenting proposed solutions in the ruling issued regarding the omission (Melchior & Courtoy, 2007). The fourth and final aspect reveals to citizens some of the aspects of Parliament's violation of their rights and freedoms and its waste of their will by not complying with its legislative obligations stipulated in the Constitutional Document, which enables them to place a vote of confidence in the members elected by them and to elect new members who submit their will and work to take care of their interests. Moreover, protect their rights, freedoms, and constitutional values (Nassar, 2011).

2.2. The effects of constitutional judicial oversight on legislative omissions regarding the quality and integrity of legislation

2.2.1. The unjustified inertia of legislative power issue

The impact of legislative negligence in the laws regulating public rights and freedoms is evident. This occurs when the legislator's regulation of these rights and freedoms is not comprehensive in its aspects and does not encompass the full dimensions of the right or liberty being regulated, which would prejudice it or diminish its effectiveness or the guarantees for benefiting from it (Salman, 2019). The violation of public rights and freedoms also occurs when the legislator remains silent about issuing legislation that regulates them for an extended period, resulting in the constitutional texts losing their value and effectiveness (Shahat, 2004; Shandy, 2017).

Therefore, the vast majority of contemporary state legislation has enshrined individual rights and freedoms at the core of the constitutional document, "the Constitution", which gives these rights and liberties the force and status of the constitutional (Sólyom, 2003). themselves Therefore. the legislator must commit not to infringe upon or detract from them once regulated legislatively. The legislator must implement the constitutional rules by enacting the necessary legislation to activate these rules (Zahra, 2013). Legislation is the primary tool for controlling a specific legislative issue in a sound legal manner, given that ensuring constitutional rights and freedoms requires their regulation by democratically elected (Mohammed, 2019).

Therefore, the ruling or decision issued by the constitutional judge declaring the existence of an unconstitutional legislative omission can be used as a means to open the door toward determining the state's responsibility, given that the legislator's failure to address the legislative omission creates a legislative gap that represents an error on the part of the legislative authority, which the courts can rely on in the judiciary to compensate the concerned parties who are harmed by this omission (El-Gamal, 2000; El-Sawy, 2003).

There is no doubt that the legislator's violation of the constitutional provisions that impose on him an obligation to legislate and do not leave him any room for freedom of discretion represents an error on his part that justifies the ruling on compensation (Csink & Pasczolay, 2008), even though the ruling unconstitutionality does not necessarily mean that

the legislator committed a mistake that requires compensation. The judge in the matter must verify the proof (Bustamante & Bustamante, 2011). The legislator made a mistake, and he is not satisfied with merely referring to the ruling of the constitutional judge (Al-Najjar, 2010).

Therefore, the constitutional judge's oversight of legislative negligence ensures that the legislator is not negligent in adopting the necessary legal provisions to guarantee and protect the rights and freedoms stipulated constitutionally by obliging the legislative authority to adopt the legislation required to protect them; otherwise, it is obligated to provide compensation to the affected parties (Al-Sanhouri, 1992).

Additionally, in the absence of a mechanism that holds the legislative authority accountable for its failure to perform its function, creating a new mechanism specifically designed to address would negligence legislative enhance the effectiveness of constitutional rules that do not permit direct application (Al-Dulaimi, 2018). In this case, parliamentary stagnation leads to an imbalance in the legal system. Therefore, the ruling or decision to add the missing legal rule is the ideal solution to restore the integrity of the legal system and achieve sovereignty for constitutional requirements, which leads to the normative enrichment of legal texts (Al-Najjar, 2010).

Hence, this oversight allows the constitutional judge to temporarily maintain, by a systematic plan, the perfection of the legal system until the legislator intervenes address to the state unconstitutionality arising from legislative negligence. This is done by adopting the new legal rule and meeting constitutional requirements (Al-Shimi, 2003). The legislator's abstention from enacting laws related to the stability of society and achieving social tranquility for prolonged periods would take the subject of legislation from the scope of appropriateness or discretion of the legislative authority to the scope of the legislative obligation to uphold the principle of the supremacy the constitution, which is the basis for monitoring the constitutionality of laws in general, and judicial constitutional oversight, especially the legislative omission (Arlettaz, 2015).

Accordingly, constitutional judicial oversight of legislative omissions enables the constitutional judge to address legal gaps in the system, ensuring respect for the rights and freedoms of constitutional value (Brewer-Carías, 2011). The judge highlights the objective deficiencies in the text subject to oversight and works to confront them to remove suspicion of its unconstitutionality and ensure the effectiveness and clarity of the law (Belgilaly, 2017b). This represents a powerful tool for confronting political authorities that exceed the limits of constitutional provisions (Hasan, 2015).

This positive intervention by the constitutional judge leads to the completion of legislative work by monitoring its quality, clarity, and comprehensiveness to ensure that the shortcomings or shortcomings in legislation that constitute a diminution or violation of public rights and freedoms or a waste of their guarantees, which it is the responsibility of the legislator to maintain their existence or increase their effectiveness, are filled and enjoy (Belgilaly, 2017a). Therefore, the constitutional courts are rushing to enhance their prominent role in constitutional judicial oversight of legislative omissions. The pioneering

role of the constitutional judge can be justified by the necessity of having a specialized body that can identify legislative omissions and the factor of time required to consider the legislator's failure to enact the overlooked legal rule as unconstitutional (Belgilaly, 2017b).

Accordingly, this constitutional judicial oversight of legislative omissions constitutes a potent tool for democratic oversight. It allows individuals to exercise a clear role in creating the legal rule if they exercise this type of constitutional oversight (Serra et al., 2008). Constitutional judicial oversight of legislative negligence requires respecting the law considering the principle of legality (Legislative omission in constitutional jurisprudence, 2008). It affirms the constitutional judiciary's competence to protect the provisions of the Constitution and preserve it from all forms of deviance by returning public bodies to their constitutional limits if they exceed them or attempt to deviate from them positively or negatively (El-Din, 1986). This is to ensure that the legislator intervenes to raise the level of protection established for rights and freedoms, as it is one of the legal mechanisms available to implement the constitutional text and activate it throughout its validity period without waiting for the legislator to intervene with a legislative procedure to establish the constitutional rule subject to implementation (Shaban, 2021).

2.2.2. Constitutional judicial oversight of legislative omission achieves legislative justice

Constitutional judicial oversight of legislative omissions is a developed judicial approach. This approach's effect is subjecting the legislator to the principle of the rule of law (Salman, 2021). This represents a significant gain for individuals, as it aims to safeguard their rights and freedoms from violence and abuse. This approach also ends a violation that, if it continues, will harm the public interest of society as a whole. Theoretical confirmation alone is insufficient to guarantee the Constitution's protection against possible attacks by Parliament (Idris, 2011). Therefore, if the legislator refrains from intervening to regulate the a particular provisions issue by the Constitution or intervenes incompletely, this means reducing the constitutional protection of the right being regulated, which constitutes a constitutional violation that must be addressed by that this legislative approach ruling unconstitutional (Azawi, 2011). What remains stable is that situations that contradict the Constitution are not subject to immunity, whether they arise intentionally or negligently, and regardless of whether they are represented by positive or negative behavior on the part of public authorities (Al-Turkmani, 2022).

Hence, if the legislator intervenes to regulate a specific legislative issue, his regulation of this issue must be accurate. The legislator's adoption of a discriminatory policy in regulating some rights seriously violates the principle of equality (Aboelazm, 2024a). This may destabilize the country's legislative negligence. system through legal Therefore, constitutional judicial oversight of this legislative omission enables the constitutional judiciary to interpret the legislative authority's work in a manner that either produces or supplements a legal rule. This is done by removing the obstacle that prevents the exercise of this right stipulated in the constitutional document "the Constitution", in addition to representing an effective tool for achieving the balance between traditional authorities and confronting their negatives by establishing a language of dialogue between the legislative and judicial authorities, which contributes to confronting the stagnation. Unjustified by the legislator (Abdullah, 2023). This is done by recognizing another authority as a mechanism to engage this stagnation through positive action, to restore balance to the legal system, by intervening to avoid the legislative omission that disturbed the balance supposed to exist in the law, and to achieve justice that the law seeks to achieve (AbdelRahman, 2016).

The meaning of equal legal protection for identical people is that it should extend to all of them, and its scope should not be limited to some of them, nor should not extend to other categories (Al-Dulaimi, 2018). Therefore, it is not permissible for this protection to be a generalization beyond its natural scope, nor for the legislator to reduce its scope by withholding it from an individual who deserves it. Through the legal texts he enacts, the legislator may intend to discriminate against the Constitution, and the effects of discrimination may prejudice the purposes of the Constitution's intent to establish it. Discrimination is considered unacceptable in both cases (Al-Murr, 2003).

Therefore, constitutional judicial oversight of legislative omissions is considered assistance to the legislator in preserving standard legislation and avoiding legislative vacuums in which it is unnecessary to declare the texts contained therein invalid or inapplicable (Mohammed, 2019). Achieving security and tranquility in any society depends on the fairness of legislation, which may fail due to the generality of the legal rule, given the difficulty of predicting its effects in advance (Mohammed, 2019).

The practical application of the legislative text may reveal the unfairness of the legal rule it contains and its violation of the principles of equality and equal opportunities (Aboelazm, 2021). Thus, the role of the constitutional judge is evident through his constitutional judicial oversight of legislative oversight, which aims to restore the legal rule to its just nature (Mohammed, 2019). This is achieved by mitigating the effects of omissions that contravene the provisions of the Constitution (Al-Dughili, 2018). Although the state and its public agencies bear responsibility for all of their actions, they must bear responsibility for their negligence in exercising their powers when this violates the rights, guarantees established and individuals, causing harm to them (Aboelazm, 2025).

In any case, despite the numerous problems surrounding constitutional judicial oversight of legislative oversight, it cannot be denied that this type of oversight facilitates progress toward implementing many of the constitutionally stipulated rules and rights (Mohammed, 2019). This ultimately leads to achieving the goal of the founding legislator regarding constitutional texts, "the Constitution", and ensuring the supremacy of constitutional rules over all other elements in the legal system. It also contributes to achieving social justice by monitoring legal texts that violate the principles of equality, equal opportunities, distributive justice, the right to litigation, fair trial, and justice in administrative, civil, and criminal procedures among members of society (Ali, 1999).

3. RESEARCH METHODOLOGY

3.1. Research methods

This paper employed a descriptive approach to determine the effects of constitutional judicial oversight on legislative omission, aiming accurately describe this phenomenon and present its integrated repercussions in an manner. encompassing all its aspects. The study also relied on the analytical approach in analyzing the rulings of the Supreme Constitutional Court, which ruled that legislative omissions were unconstitutional, to determine the effects that these rulings have on individual rights and freedoms and on the principle of separation of powers and the extent of the intervention of the constitutional judiciary in the process of making legislation. In addition, the analytical approach has helped to determine the effects of the rulings of the Supreme Constitutional Court on the quality and integrity of legislation, the impact of this type of oversight on the relationship of the constitutional judiciary with the legislative authority, and how exercising jurisdiction over constitutional judicial oversight over legislative omission can affect the jurisdiction of the Supreme Constitutional Court.

3.2. Data collection

This paper relies on several sources, the most important of which are the rulings issued by the Constitutional Court, to analyze them and understand the effects that result from these rulings, in addition to the laws regulating the work of the Supreme Constitutional Court and its jurisdiction under the law. The Constitution was also relied upon as one of the sources of data and information to determine the framework. The Constitutional Court governs and regulates the work of the Supreme Constitutional Court. Several books in Arabic and English were also consulted, as were several scientific research articles published in international journals by prominent publishers such as Elsevier, Emérald, Sage, Interscience, and other reputable international publishing houses. These studies have also been published in peer-reviewed scientific journals indexed in international databases, including Scopus, Web of Science (WoS), Google Scholar, ResearchGate, and ABDC.

4. RESULTS

4.1. The relationship between the judicial and legislative authorities

4.1.1. Constitutional judicial review and the relationship with the legislative authority

There is an inverse relationship between the activity and effectiveness of constitutional courts and the level of their independence from parliaments. The more active and extensive the court was in issuing unconstitutional rulings. more the extraordinary Parliament's hostility towards it (Haji, 2012). Primarily, the legislation under supervision reflects certain balances that the legislator aims to protect through agreed-upon interests (Al-Turkmani, 2022). Despite the deeply rooted doctrine of the constitutional judiciary, which

is the necessity of respecting the legislator's will, constitutional oversight sometimes leads to profound interference in exercising the legislative function (Blondio-Mondoloni, 2014).

Hence, the intervention of the constitutional oversight body to impose its oversight on the work of the legislative authority leads to its inconvenience (Bustamante & Bustamante, 2011). This is considered to be the court stigmatizes its actions as violating the Constitution for the purposes it should protect. This stigmatizes members of the legislative authority by violating their constitutional oath, which they swore to uphold the Constitution, especially if the legislative text does not violate the Constitution's provisions, as in the case of legislative omission (El-Gamal, 2000; Csink & Pasczolay, 2008).

Moreover, exercising jurisdiction through constitutional judicial oversight over legislative omissions is an exceptional tool that constitutes an unwanted intrusion into the jurisdiction of that authority (Brewer-Carías, 2011). However, given that the constitutional courts are trying to blur the difference between the legislative function and the judicial function, constitutional judicial oversight of legislative omission revives the traditional theory of the "principle of separation of powers" through the legal effects of decisions or rulings declaring the existence of legislative omission, whether the omission arises from legislative action or administrative (Al-Ghafloul, 2003). Especially when considering the phenomenon of judicial activity in constitutional courts. The constitutional judge intends to interfere in societal policy-making, bypassing his primary task of constitutional oversight (Azawi, Al-Sanhouri, 1992).

Therefore, the intervention of the constitutional judge in the legislative function of the legislative and executive branches leads to the merging of the various state functions and a dangerous concentration of power in the hands of a single body, "the constitutional judiciary", that is not subject to any fundamental legal oversight, whether by other authorities or by the electorate, which leads to undermining the supervisory role of the judicial authority Itself (Al-Shennawy, 2017). Although constitutional judicial control over legislative negligence is appropriate for countries with stable democracies, it can become a tool that usurps the tasks or functions of legislative bodies in newly established democracies (Al-Dughili, 2018). This creates permanent tension between the constitutional judiciary and the legislative authority (Al-Murr, 2003).

4.1.2. The legislation-making process

Constitutional judicial oversight of legislative omissions raises concerns about the potential drift toward overlap between legislative and judicial functions, particularly when the rules of the democratic game are unstable within the state's legal system (Al-Najjar, 2010). This oversight could become a pretext for judicial control over the legislative function, leading to constitutional courts determining public policies and exposing the legal system to the risk of politicizing the constitutional judiciary (Belgilaly, 2017a).

Unsurprisingly, constitutional judicial oversight of legislative negligence often leads to excessive judicial activity, which manifests in rulings or decisions that supplement or interpret legislative texts (Belgilaly, 2017b). This leads to an overlap between the judicial function of the constitutional courts and the legislative function of Parliament about existing legislation (Legislative omission in constitutional jurisprudence, 2008). It leads to going beyond that by sometimes declaring new legislative rules or deriving them from the texts and spirit of the Constitution (Melchior & Courtoy, 2007).

The distinctive characteristics of these rulings or decisions highlight the sensitivity of the role played by constitutional courts in filling the normative content gap that affects the text subject to their control, whether narrowing or expanding (Mohammed, 2019). No matter how many methodological arguments are presented to justify constitutional judicial oversight of legislative negligence, it is impossible to hide the growing role of the constitutional judge once they address the defects in the legal system arising from a lack of legislative activity (Shahat, 2004).

It is also unacceptable to have a role in which the constitutional judiciary becomes a positive legislator, resulting in legal insecurity (Sólyom, 2003). It considers the phenomenon of exaggerated activity of the constitutional judiciary or excessive politicization of its rulings and decisions. Using constitutional judicial oversight of legislative omissions to include in the text subject to oversight a legal rule that is intolerable or content that is unacceptable, exceeds the limits of the legislative will, which represents exceeding the limits of supervision of the constitutionality of laws, and a substitution of the constitutional judiciary for Parliament in exercising its legislative function (Zahra, 2013).

Accordingly, the constitutional and democratic caveats and risks surrounding the interference of the constitutional judiciary in the legislative process, whether direct or indirect, cannot be overlooked or ignored (Shaban, 2021). The involvement of constitutional courts in establishing law and shaping legislative policies has several disadvantages (El-Sawy, 2003). The most prominent of these negatives is the inability of constitutional courts to envision the far-reaching consequences and effects of their decisions on the broader societal landscape, given their limited capacity to gather information (El-Din, 1986).

Recognizing the normative authority the constitutional judge also represents a significant threat to the balance of powers, given the finality of his rulings and decisions and the impermissibility of appeal against them (Haji, 2012). Instead of constitutional judicial oversight of legislative negligence becoming a tool for legal security, it becomes a source of anxiety or tension for the legislative authority, which may lead it to retreat from carrying out its legislative role or carry it out in a defective manner (Arlettaz, 2015). This may lead them to take the initiative to the constitutional provisions of their content by getting rid of their effects under original or subsidiary legislation or reducing the powers of the oversight body in a way that limits the effectiveness of its role, even if the matter requires Parliament to amend the constitution, to end the conflict between it (Al-Aboudi, 2010) and the constitutional judiciary. This is achieved by creating a mechanism to address the effects of constitutional provisions that negatively impact the legal system, particularly individual rights and freedoms (Abdullah, 2023).

Constitutional judicial oversight of legislative negligence leads to a reevaluation of the political alternative chosen by the legislator, resulting in a legislative vacuum that affects the integrity of the legal system (Abdulkareem & Al-Amarah, 2019). judge the constitutional exercises comprehensive control over the legislative activity of the legislator, which exposes him to many criticisms related to the government of judges. The judge is the only one whom the legislator cannot deviate from what he decides, given the specificity of the finality of the rulings or decisions he issues and the impermissibility of appealing them, which leads to the possibility that the constitutional judge poses a severe threat to the balance between powers, as he is considered the guarantor of this balance (Abdel-Badi, 2019).

Therefore, rulings or decisions issued within the scope of constitutional judicial oversight over legislative omissions may undermine the principle of separation of powers. Given the high financial cost that these provisions and decisions entail, they represent a violation of Parliament's exclusive jurisdiction to approve the public budget (Abboud, 2019). Suppose the constitutionally stipulated rights positive and freedoms require legislative intervention to guarantee and protect them (Aboelazm, 2021). In that case, this intervention is limited by the state's financial capacity. Therefore, estimating the limits of this intervention should be left to the legislative authority alone, especially since the constitutional judge lacks sufficient information about the public financial burdens (Al-Dulaimi, 2018). The state may not have enough funding to implement the ruling of the constitutional judge in cases of legislative omission. This issue is considered extremely sensitive, especially when considering the constitutional judiciary's tendency to condemn the state's negligence in implementing Fundamental rights and to burden it with proving the lack of state financial resources (Aboelazm, 2024b).

Therefore, the Federal Constitutional Court in Germany has done well in following a rational judicial policy regarding unconstitutional legislative omissions in financial laws, such as taxes, social insurance, wages, and salaries (Abu Halima, 2015). It left room for the legislative authority to intervene with the necessary legislation to fulfill constitutional requirements within the framework of available state resources (AbdelRahman, 2016).

4.1.3. Constitutional judicial oversight of legislative omission leads to usurping the power of legislation

A constitutional judge may use constitutional judicial oversight of legislative omissions to determine some benefit for himself. This is done by extending the legislator's benefits to others, including himself, violating the legislator's will (AbdelRahman, 2016). It must not be forgotten that constitutional courts may use constitutional judicial oversight of legislative omissions to usurp the legislator's authority, which constitutes a deviation from judicial oversight of the constitutionality of laws (Al-Dulaimi, 2018). Therefore, the legitimacy of oversight remains dependent constitutional judge's respect for the will of the legislator that he seeks from the legislative text so that the restriction or rule guaranteed by the constitutional judge in this text remains covered by the will of the legislator and operates within its

framework (Al-Dulaimi, 2018). Therefore, every restriction or rule that does not respect the content of the legislative text as intended by the legislator and which results in the continuation of this text, in reality, represents the continuation of this text in the legal system against the will of the person with inherent jurisdiction over legislation, "Parliament", which means a standard amendment to the judicial rulings in the text (Al-Janabi, 2022). Subject to censorship is an exert case of a legislative not mandate that is permissible a constitutional standpoint (Haji, 2012).

The bottom line is that the constitutional judge's resort to constitutional judicial oversight over legislative negligence without having a close and direct connection to his primary mission of monitoring constitutionality turns this oversight into a reason or loophole for escaping the reins of the Constitution, leading to aggression against Parliament's legislative jurisdiction (Aboelazm, 2025).

4.2. The effects of constitutional judicial oversight on the legislative omission regarding the jurisdiction of the constitutional judiciary

4.2.1. Constitutional judicial oversight of legislative omissions weakens confidence in the rulings of the constitutional judiciary

It is challenging for state agencies to regulate themselves once they exercise the authority assigned to them by the Constitution (Al-Dulaimi, 2018). Hence, constitutional courts may abuse their constitutional jurisdiction without citizens or other constitutional and regulatory bodies being able to monitor their work. Thus, constitutional judicial oversight of legislative oversight turns into an unconstitutional or illegitimate tool (Al-Dulaimi, 2018). This occurs when constitutional courts are readily amenable to government influence. It usurps the legislator's authority and function, or does something worse by illegally altering the provisions of the Constitution itself (Ali, 1999).

If the Constitutional Court submits to the will of the political authority, it will transform from a protector of the Constitution's provisions into a more authoritarian tool. Considering that the occurrence of the best system of constitutional justice in the hands of a judge subject to political authority represents a death certificate for individuals' freedoms and a tool to circumvent the provisions of the Constitution (Al-Sayed, 2009), which leads to the loss of individuals' confidence in the rulings and decisions of the constitutional judiciary (Al-Shimi, 2003).

Likewise, the constitutional judge's exploitation of the constitutional judicial oversight of the legislative omissions to stand up to the public authorities and impose guardianship over them turns this oversight into a tool for the destruction of sound democratic life due to the political and constitutional problems that this approach leads to between the oversight body and the public authorities, and the possibility of the latter encroaching on them (Csink & Pasczolay, 2008). This may threaten its independence or portend the aggression of public authorities against the principle of legality and the collapse of the legal fence that protects the rights and freedoms of individuals (Al-Ghafloul, 2003).

Constitutional courts are not immune to pressure. Whenever it resorts to establishing general legal rules through constitutional judicial oversight of legislative omissions, disagreement and political controversy arise over the content of these rules (Al-Turkmani, 2022). Therefore, resolving this political dispute will not be limited to settling the conflict at hand. Instead, it may extend to undermining the powers of the constitutional judiciary as a whole through legislative amendments, whose aim is to limit the powers of the constitutional courts under the pretext of reforming the constitutional justice system, thus affecting the level of protection of fundamental rights (Belgilaly, 2017b).

Also, the use of constitutional courts for constitutional judicial oversight of legislative omissions is a means of usurping the powers of other authorities and exceeding the scope of their jurisdiction to carry out constitutional oversight by amending the effects of their rulings or applying them to the impact of the verdict issued as unconstitutional by themselves to the case presented before the judge of the matter, which weakens confidence in the rulings of the constitutional judiciary (Aboelazm, 2021; Brewer-Carías, 2011). This leads to an attempt by ordinary courts to evade the ruling on the constitutional issue entirely and to dispute the centrality of constitutional oversight once again (Blondio-Mondoloni, 2014).

It is established that the trial court has the authority to assess the impact of the ruling of unconstitutionality on the substantive case before it. Any attempt by others to explain this impact is considered an infringement on their inherent jurisdiction (Bustamante & Bustamante, 2011). Therefore, when considering the preliminary issue, the constitutional judge may not address the settlement of the substantive dispute nor accept or reject any substantive defense presented to him by the opponents. The constitutional lawsuit is not an objective dispute (Melchior & Courtoy, 2007). Still, the scope is limited to deciding on a constitutional dispute regarding the texts governing the objective dispute (Idris, 2011).

Accordingly, the jurisdiction of the Supreme Constitutional Court does not focus on determining the legal rule applicable to the substantive dispute. Instead, it focuses only on the constitutional issue raised regarding this dispute (Hasan, 2015). The ruling issued by it is not authoritative in this regard, but rather is authoritative regarding the constitutionality or unconstitutionality rule governing substantive the legal the disagreement. Explaining the effect of the judiciary's unconstitutionality in applying it to the facts of the dispute presented is within the jurisdiction of the court of subject matter to which the dispute is given alone and is outside the scope the jurisdiction of the constitutional (Halmai, 2019).

4.2.2. Undermining the powers of the constitutional judiciary

The intervention of the constitutional judge in creating a legal rule and formulating legislative policies leads to these rules and policies escaping the constitutional oversight exercised by the constitutional oversight body. No authority in

the state enables it to review the constitutional work of lawyers. The constitutional judiciary is the only body whose work is not subject to oversight, examination, or review (Al-Khalaileh et al., 2024; El-Sawy, 2003). Therefore, his abuse of his position falls outside the scope of enforcing constitutional provisions. Although it is conceivable the requirements of the Constitution would be violated by other state bodies, such as the parliament, it is unacceptable for the constitutional iudge to violate the provisions of the Constitution that he is implementing and protecting to impose his absolute control over other authorities, under the pretext of being the protector of constitutional rules. Constitutional judicial oversight of legislative omissions, like other forms of judicial oversight, can be subject to deviation from its correct course without citizens or other constitutional authorities having the means to confront this deviation (Belgilaly, 2017a).

The constitutional judge may use this type of oversight to expand his powers beyond those stipulated by the Constitution and law (Al-Shimi, 2003). This is undesirable because it constitutes a usurpation of the authority of the founding legislator himself (Azawi, 2011). The motivation of the constitutional courts in seeking constitutional judicial oversight of legislative omissions may not be to respond to the constitutional violation as much as their desire to have the highest say in critical legislative matters. This raises the ire the legislative authority towards it, so it intends to attack the constitutional jurisdiction assigned to it (Serra et al., 2008). This is achieved by restricting this jurisdiction legislatively, either by depriving the constitutional oversight body of explicit constitutional judicial oversight over legislative omissions, by not granting it the right to constitutional oversight altogether, or by imposing conditions that are difficult to meet (Mohammed, Consequently, constitutional oversight is based on legislative negligence, harm to the legal system, and a waste of the constitutional guarantees established for rights and freedoms. Therefore, caution must be exercised undertaking this control (Salman, 2021).

Constitutional judicial oversight of legislative omissions by constitutional courts goes beyond the traditional limits of constitutional oversight. Experience has shown that many decisions and rulings of the constitutional judiciary, issued to address legislative omissions, exceed the limits and restrictions of this type of constitutional oversight (Salman, 2019). The constitutional text may reduce the effects of the decision or ruling to address legislative oversight, limiting it to merely communicating with the overlooked authority (legislative or executive) without forcing it to intervene with legislation, or at the most, it sets a final date for enacting the legal rule with the knowledge of that authority, without deciding a more mandatory legal outcome (Salman, 2021). The text of the Constitution does not specify the penalty that a judge can impose on the negligent "legislative or executive" authority if it does not intervene in establishing the legal rule that was ignored within the period stipulated by the Constitution. He is satisfied with the role of the constitutional judge in revealing legislative oversights without intervening to address them himself (Shandy, 2017). However, the constitutional judge does not stop there and decides to address

the legislative omissions nonetheless, which affects the relationship of the constitutional oversight body with other authorities (Shaban, 2021).

Additionally, the constitutional attempts to exercise constitutional judicial oversight of legislative omissions to challenge reform programs and impose their guardianship and methods on legislative authority inevitably raise sensitivity between them and the public authorities in the state (Sólyom, 2003). Therefore, when a constitutional judge issues a standard ruling or decision to address a legislative omission, the constitutional judge must take the necessary precautions to limit the effects of this ruling or decision over time (Zahra, 2013; Shahat, 2004). The constitutional judge can address a specific case without a legal rule, provided that the impact of the verdict or decision is temporary and directly related to the subject of the case, and not to any other matter (Legislative omission in constitutional jurisprudence, 2008).

The constitutional judge's establishment of legislative rules that have far-reaching political repercussions through rulings issued the scope of constitutional judicial oversight of legislative omissions leads to the political authority resorting to several means to dissolve the judicial concepts established by the constitutional judiciary or to restrict it (El-Gamal, 2000). The political authority can limit the constitutional judiciary, such as organizing campaigns to dismiss constitutional influencing the formation judges, constitutional courts through the appointment process, mobilizing public opinion, organizing demonstrations, and encouraging resistance against the expansionist policy of the constitutional courts, in a way that undermines the jurisdiction of the constitutional judiciary in the end (El-Din, 1986; Haji, 2012).

5. CONCLUSION

This paper examines the impact of constitutional oversight on the omissions the constitutional protection of rights and freedoms. This aims to strengthen the principle of legal achieve security and integration hetween the functions of the constitutional judiciary and those of public authorities. It also added a new dimension to the established presumption of constitutionality of legislation. The paper also addressed the effects of constitutional judicial oversight on legislative omissions regarding the quality and integrity of legislation, to address the unjustified inertia of the legislative authority. The impact of judicial review of legislative omissions on achieving legislative justice was also analyzed.

Furthermore, the impact of constitutional on the relationship between constitutional judiciary and the legislative authority was addressed. This type of review may lead to the relationship hetween tension in constitutional judiciary and the legislative authority, threatening the entire legislative process. This may lead to the usurpation of legislative authority, which weakens confidence in the rulings of constitutional court and undermines its powers.

This study reached several results, namely the importance of judicial oversight of legislative omissions and the positive effects and results it can achieve, namely strengthening the principle of legal security, since if Parliament neglects or neglects to protect some rights and freedoms, the Supreme Constitutional Court, by its authority to monitor the constitutional Despite the legislative oversight, it can address this, which achieves legal security for individuals in society. In addition, constitutional judicial oversight of legislative oversight would give the Supreme Constitutional Court the authority and jurisdiction to address any shortcomings emanating from the legislative authority, which achieves integration between the judicial authority represented by the Supreme Constitutional Court and the legislative authority represented by Parliament, for the benefit of individuals.

In addition, constitutional judicial oversight of legislative omissions is considered one of the tools possessed by the Supreme Constitutional Court, adding one of the new dimensions to constitutional oversight of legislation, determined to be in the interest of legislation in terms of the integrity of legislation, in addition to avoiding the stagnation of legislative authority. However, constitutional judicial oversight of legislative omissions has some adverse effects, which are represented by the tension in the relationship between the constitutional judiciary and the legislative authority. It may lead to jeopardizing the powers of the constitutional judicial oversight of legislative omissions can also help in legislative despotism and the constitutional judiciary's usurpation of legislative authority.

REFERENCES

Abboud, S. A. (2019). Aimtinae alsultat altashrieiat ean 'ada' wazifatiha altashrieia [Refusal of the legislative authority to do its job] [Master's thesis, Alexandria University]. Alexandria University.

Abdel-Badi, S. M. S. (2019). Hudud raqabat alqadi aldusturii ealaa alsultat altaqdiriat lilmusharie (dirasat muqaranatin) [The limits of the constitutional judge's review of the legislator's discretionary power: (comparative study)]. Dar Al-Nahda Al-Arabiya.

AbdelRahman, G. A. (2016). Alraqabat aldusturiat ealaa al'iighfal altashrieii: Dirasat tahliliat muqaranatan [The judicial review on the legislative omission: An analytical comparative study]. Dar Al-Nahda Al-Arabia.

Abdulkareem, S. H., & Al-Amarah, H. T. (2019). Athar al'iighfal altashrieii aliajtimaeia [The social influences of legislative omission: A comparative study]. Journal of Legal Sciences, 34(2). https://jols.uobaghdad.edu.iq /index.php/jols/article/view/234/195

Abdullah, F. Y. (2023). Mawqif almahkamat aldusturiat fi alkuayt min alraqabat ealaa dusturiat al'iighfal altashrieii "dirasat tasiliat — tahliliat — muqaranatun [The situation of the constitutional court in Kuwait regarding the oversight of the constitutionality of legislative omission: A rooting analytical comparative study]. Journal of Law for Legal and Economic Research, 1(1).

Aboelazm, K. S. (2021). The constitutional framework for public policy in the Middle East and North Africa countries.

**International Journal of Public Law and Policy. 7(3), 187-203. https://doi.org/10.1504/IJPLAP.2021.118325

**Aboelazm, K. S. (2024a). Supreme constitutional court review of the legislative omission in Egypt of Public Law and Policy. 7(3), 187-203. https://doi.org/10.1504/IJPLAP.2021.118325

international experiences. Heliyon, 10(17), Article 37269. https://doi.org/10.1016/j.heliyon.2024.e37269

- Aboelazm, K. S. (2024b). The role of judicial review on the settlement of state contracts disputes. Corporate Law & Governance Review, 6(3), 122-134. https://doi.org/10.22495/clgrv6i3p13
- Aboelazm, K. S. (2025). The judicial constitutional review for the legislative omission: A comparative study. Krytyka Prawa. Niezalezne Studia nad Prawem, 17(1), 316-336. https://doi.org/10.7206/kp.2080-1084.766
- Abu Halima, E. M. (2015). Alraqabat alqadayiyat ealaa alsultat altaqdiriat lilmusharae dirasat muqarana [Judicial review over the discretionary power of the legislator: A comparative studyl. Dar Al-Nahda Al-Arabiya.
- Al Dalaien, A. A., & Aladaseen, M. A. (2025). Judicial mediation and the governance of administrative contract disputes: The legal possibilities under Jordanian and UAE legislation. *Corporate Law & Governance Review,* 7(1), 111-122. https://doi.org/10.22495/clgrv7i1p10
- Al-Aboudi, M. (2010). Altadabir al'amniat waneikasatuha ealaa alhuriyaat alshakhsiat lil'afrad [Security measures and
- their repercussions on the personal freedoms of individuals]. Security Information Center, Police College. Al-Dughili, S. F. (2018). *Al'iighfal altashrieiu fi alqada' aldusturii* [Legislative omission in the constitutional judiciary]. Mauritanian Journal of Law and Economics, 25.
- Al-Dulaimi, A. (2018). From negative to positive legislator? Response to unconstitutional legislative omission as a case study in the changing roles of constitutional courts [Doctoral thesis, Griffith University]. Griffith University. https://research-repository.griffith.edu.au/server/api/core/bitstreams/d4d9c497-ae90-42f4-8db0-55b5565b2d3e/content
- Al-Ghafloul, E. A. (2003). Fikrat eadam aliakhtisas alsalbii lilmusharie (dirasat muqaranatin) [The idea of lack of negative jurisdiction for the legislator: Comparative study] (2nd ed.). Dar Al-Nahda Al-Arabiya.
- Ali, S. E. (1999). Haqiqat alfasl bayn alsulutat fi alnizam alsiyasii waldusturii lilwilayat almutahidat al'amrikia [The fact of the separation of powers in the political and constitutional system in the United States of Americal [Doctoral thesis, Ain Shams University]. Ain Shams University.
- Al-Janabi, M. F. (2022). Raqabat alqadi aldusturii ealaa alaimtinae altashrieii [The constitutional judge's review of legislative abstinence]. Dar Al-Nahda Al-Arabiya.
- Al-Khalaileh, L., Al-Billeh, T., Al-Oheiw, M. A., & Almamari, A. (2024). Legal controls for discounting commercial papers in Jordanian legislation and its judicial applications legislation. Corporate Law & Governance Review, 6(4), 53-62. https://doi.org/10.22495/clgrv6i4p5
- Al-Murr, A. (2003). Alraqabat alqadayiyat ealaa dusturiat alqawanin wallawayih fi malamihiha alrayiysia [Judicial review of the constitutionality of laws and regulations in their main features]. Rene Jean Debuy Center for Law and Development.
- Al-Najjar, M. E. (2010). Fi alqada' bieadam dusturiat alaghfal altashrieii (qada' eadam aldusturiat alshurti) [In judging the unconstitutionality of legislative omission: Judicial unconstitutionality conditional]. Al-Dusturiyyah Magazine, 17.
- Al-Sanhouri, A. A. (1992). Mukhalafat altashrie lildustur walanhiraf fi astiemal alsultat altashrieia [Violation of the legislation to the constitution and deviation in the use of the legislative authority]. Journal of Law and Economics.
- Al-Sayed, A. H. (2009). Dawr alqadi aldusturii fi alraqabat ealaa alsultat altaqdiriat lilmusharae Dirasat muqarana [The role of the constitutional judge in supervising the discretionary power of the legislator: A comparative study]. Dar Al-Nahda Al-Arabiya.
- Al-Shennawy, W. M. (2017). Dawr almahakim aldusturiat kamusharie 'iijabiin (dirasat tasiliat tahliliat muqaranatin) [The role of constitutional courts as positive legislators: A comparative analytical originality study]. Journal of Legal and Economic Research, 62.
- Al-Shimi, A. A (2003). Raqabat al'iighfal altashrieii fi qada' almahkamat aldusturiat aleulya Dirasat muqarana [The review of legislative omission from the supreme constitutional court: A comparative study] (3rd ed.). Dar Al-Nahda Al-Arabiya.
- Al-Turkmani, O. H. (2022). Alakhtisas alraqabiu lilmahkamat aldusturiat alfilastiniat ealaa aliamtinae altashrieii "dirasat tahliliat muqaranatin" [The supervisory jurisdiction of the Palestinian constitutional court over legislative abstention: A comparative analytical study]. *Journal of Sharia and Law, 91*(91), Article 5. https://digitalcommons.aaru.edu.jo/cgi/viewcontent.cgi?article=1004&context=sharia_and_law
- Arlettaz, J. (2015). L'incompétence négative à l'étranger [Negative incompetence abroad]. Les nouveaux Cahiers du Conseil Constitutionnel, 46, 55-75. https://doi.org/10.3917/nccc1.046.0055
- Azawi, A. A (2011). Dawabit tawzie alaikhtisas bayn alsultatayn altashrieiat waltanfidhia (dirasat muqaranat fi tahdid majal kulin min alqanun wallaayihat fi alnizam alsiyasii aljazayirii) — aljuz' al'awal [Criteria for the distribution of competence between the legislative and executive authorities (A comparative study in defining the scope of each of the laws and regulations in the Algerian political system) — Part one]. Dar Al-Gharb for Publishing and Distribution.
- Belgilaly, K. (2017). Aliakhtisas alsalbiu lilmusharie walraqabat aldusturiat ealayh dirasat muqarana [The negative jurisdiction of the legislator and constitutional review on it: A comparative study]. Journal of Al-Ostaz Al-Baheth for Legal and Political Sciences, 6.
- Belgilaly, K. (2017b). *Alsultat altaqdiriat lilmusharae dirasat muqarana* [The discretionary power of the legislator: A comparative study] [Doctoral thesis, Abi Bakr Belkaid University]. Abi Bakr Belkaid University.
- Blondio-Mondoloni, V. (2014). Finances publiques et droits fondamentaux: Essai sur les relations entre les finances publiques et les droits fondamentaux (Public finances and fundamental rights: Essay on the relationship between public finances and fundamental rights) [Doctoral thesis, Université d'Aix-Marsille]. Université d'Aix-Marsille. https://theses.fr/2014AIXM1098
- Brewer-Carías, A. R. (2011). Constitutional courts as positive legislators in comparative law, in constitutional courts as positive legislators: A comparative law study. Cambridge University Press.
- Bustamante, T., & Bustamante, E. D. G. (2011). Constitutional courts as negative/positive legislators: The Brazilian case. In A. Brewer-Carías (Ed.), Constitutional courts as negative legislators (pp. 283-314). Cambridge University Press. https://doi.org/10.1017/CBO9780511994760.013
- Csink, L. & Pasczolay, P. (2008). Problems of legislative ommission in constitutional jurisprudence [Hungarian national report presentation]. The 14th Conference of Constitutional Courts https://www.confeuconstco.org /reports/rep-xiv/report_Hungary_en.pdf?
- El-Din, S. G. (1986). Lawayih aldarurat wadamanat alraqabat alqadayiya [Necessity regulations and guarantees of judicial review]. Munsha'at Almaearif.
- El-Gamal, Y. (2000). Alqada' aldusturiu fi misr [The constitutional judiciary in Egypt]. Dar Al-Nahda Al-Arabiya.

- El-Sawy, A. (2003). Alsiyaghat altashriciat lilhukm aljayidi? 'iitar muqtarah lilduwal alearabia [Legislative formulation for good governance? A proposed framework for Arab countries] [Paper presentation]. Workshop on Developing a Model for Legislative Drafting for Arab Parliaments, Beirut, Lebanon.
- Haji, G. M. (2012). Alsiyasat alqadayiyat lilmahkamat aldusturiat aleulya Dirasat muqarana [The judicial policy for the constitutional supreme court: A comparative study] (1st ed.). Dar Al-Nahda Al-Arabia.
- Halmai, G. (2019). Dismantling constitutional review in Hungary. *Rivista Di Diritti Comparati,* 1. https://www.diritticomparati.it/wp-content/uploads/2019/07/2-Halmai.pdf
- Hasan, H. M. (2015). Muealajat aimtinae albarlaman ean mubasharat aikhtisasih altashrieii fi alqanun alwadeii [Processing parliament's refusal to practice its legislative jurisdiction in positive law]. *Journal of Al-Mohaqiq El-Haly for Legal and Political Sciences*, 7(4).
- Idris, A. A. (2011). *Raqabat almula'amat fi alqada' aldusturi: Dirasat muqarana* [Suitability review in the constitutional judiciary: A comparative study] (1st ed.). Dar Al-Nahda Al-Arabiya.
- L'omissione legislativa nella giurisprudenza costituzionale [Legislative omission in constitutional jurisprudence]. (2008). [Report presentation]. The 14th Congress of the Conference of European Constitutional Courts. https://www.confeuconstco.org/reports/rep-xiv/Legislative_Omission_Constitutional_Jurisprudence_it.pdf Melchior, M., & Courtoy, C. (2007, December 4). L'omission législative dans la jurisprudence constitutionnelle
- Melchior, M., & Courtoy, C. (2007, December 4). L'omission législative dans la jurisprudence constitutionnelle [Legislative omission in constitutional jurisprudence]. Cour constitutionnelle de Belgique. https://www.confeuconstco.org/reports/rep-xiv/report_Belgium_fr.pdf
- Mohammed, A. Q. M. (2019). The monitoring of legislative omission by the constitutional judiciary: A comparative study. [Doctoral thesis, University of Reading]. University of Reading. https://centaur.reading.ac.uk/88724/
- Nassar, G. (2011). Al'ada' altashrieiu limajlis alshaeb walraqabat ealaa dusturiat alqawanin fi misr [Legislative performance of the People's Assembly and judicial review of the constitutionality of laws in Egypt]. Dar Al-Nahda Al-Arabia.
- Salman, A. M. (2021). Alraqabat alqadayiyat ealaa qusur altanzim altashrieii: Dirasat fiqhiat qadayiyat mudaeamat bijamie ma 'asdarath almahkama aldusturiat aleulya min 'ahkam [Judicial review of the deficiencies of legislative regulation: A jurisprudential study supported by all the rulings issued by the Supreme Constitutional Court] (1st ed.). The National Center for Legal Publications.
- Salman, S. D. (2019). Al'iighfal altashrieiu wa'iimkaniat fard alraqabat alqadayiyat ealayh fi aleiraq [Legislative omission and the possibility of imposing judicial review on it in Iraq]. *Journal of the Faculty of Law, 21*(4).
- Serra, B., de Almeida, L. N., Maurício, A., da Fonseca, G., dos Prazeres Pizarro Beleza, M., de Sousa e Brito, J., Brito, M. H., Palma, M. F., Tavares da Costa, A., Mota Pinto, P., & Cardoso da Costa, J. M. (2008). *Portuguese Report for the XIVth Congress of the Conference of European Constitutional Courts.* Portuguese Constitutional Court https://www.confeuconstco.org/reports/rep-xiv/report Portuguese%20 en.pdf
- Constitutional Court. https://www.confeuconstco.org/reports/rep-xiv/report_Portuguese%20_en.pdf Shaban, R. D. (2021). *Raqabat alqada' aldusturii ealaa al'iighfal altashrieii "dirasat muqarana"* [The judicial review on the legislative omission: A comparative study]. [Doctoral thesis, Helwan University]. Helwan University.
- Shahat, H. F. (2004). *Alhimayat aldusturiat lilhaqi fi almusawat Dirasat muqarana* [The constitutional protection for the right of equality: A comparative study]. [Doctoral thesis, Cairo University]. Cairo University.
- Shandy, S. (2017, October 8–9). Sulutat alqadi al'iidarii fi muajahat wasadi alnaqs altashrieii fi alqanun al'iidarii [The administrative judge's authorities in confronting and filling the legislative deficiency in administrative law] (Working paper submitted to the second scientific forum of the Arab Union for Administrative Judiciary under the title "The administrative judge's authorities in directing the administration and filling the shortage of legislative"). Arab Federation of Administrative Judiciary.
- Sólyom, L. (2003). The role of constitutional courts in the transition to democracy: With special reference to Hungary. *International Sociology*, 18(1), 133–161. https://doi.org/10.1177/0268580903018001008
- Zahra, K. (2013). *Alaghfal altashrieiu walriqabat aldusturiat ealayh* [Legislative omission and constitutional oversight on it] [Master's thesis, Abi Bakr Belkaid University]. Abi Bakr Belkaid University.